Author’s response to reviews

Title: The study protocol of: ‘Initiating end of life care in stroke: Clinical decision-making around prognosis’

Authors:

Christopher R Burton (c.burton@bangor.ac.uk)
Sheila Payne (s.a.payne@lancaster.ac.uk)
Mary Turner (j.m.turner@lancaster.ac.uk)
Tracey Bucknall (tracey.bucknall@deakin.edu.au)
Jo Rycroft Malone (j.rycroft-malone@bangor.ac.uk)
Pippa Tyrrell (pippa.tyrrell@manchester.ac.uk)
Marie Horne (m.horne@bradford.ac.uk)
Lupetu Ives Ntambwe (l.i.ntambwe@bangor.ac.uk)
Sarah Tyson (sarah.tyson@manchester.ac.uk)
Helen Mitchell (Helen.mitchell@wales.nhs.uk)
Sion Williams (sion.williams@bangor.ac.uk)
Salah Elghenzai (salah.elghenzai@wales.nhs.uk)

Version: 2  Date: 20 November 2014

Author's response to reviews:

I have corrected the spelling of one of the co-authors' names.